Author’s response to reviews

Title: Screening for type 2 diabetes in a high-risk population: study design and feasibility results of a population-based randomized controlled trial

Authors:

Bart Klijs (b.klijs@erasusmc.nl)
Suzie J Otto (s.otto@erasusmc.nl)
Robert J Heine (RJ.Heine@vumc.nl)
Yolanda van der Graaf (Y.vanderGraaf@umcutrecht.nl)
Jan J Lous (j.j.lous@star-mdc.nl)
Harry J de Koning (h.dekoning@erasusmc.nl)

Version: 2 Date: 29 December 2011

Author’s response to reviews:

Dear Editor,

We gratefully acknowledge the opportunity given by the journal for considering a revised manuscript of our paper.

The comment given by the Associate Editor is evaluated and the necessary changes were made in the Introduction section of the manuscript and highlighted in the uploaded version.

We feel that with the revisions the novelty of our paper is clear now and our manuscript suitable to enter the review process.

Sincerely yours,
on behalf of all authors,

Suzie J. Otto, Ph.D.